**Author's response to reviews**

**Title:** Patch: platelet transfusion in cerebral haemorrhage: study protocol for a multicentre, randomised, controlled trial.

**Authors:**

Koen de Gans (k.degans@amc.uva.nl)
Rob J de Haan (R.J.deHaan@amc.uva.nl)
Charles B Majoie (c.b.majoie@amc.uva.nl)
Maria M Koopman (r.koopman@sanquin.nl)
Anneke Brand (a.brand@sanquin.nl)
Marcel G Dijkgraaf (m.g.dijkgraaf@amc.uva.nl)
Marinus Vermeulen (m.vermeulen@amc.uva.nl)
Yvo B Roos (y.b.roos@amc.uva.nl)

**Version:** 2  **Date:** 13 March 2010

**Author's response to reviews:**

Major revisions:
- The author list is written exactly as they are in the submission system.
- The abstract structure has been reformatted.
- The manuscript structure has been reformatted.
- An ethical considerations paragraph has been added to the Methods/Design section of the manuscript.

Minor revisions:
- The text title: has been removed from above the title.
- The title has been reformatted.
- The authors’ contributions has been formatted into a paragraph with initials to refer to each author.
- The appendices have been moved to before the acknowledgements section.